Author's response to reviews

Title: Using theatre in education in a traditional lecture oriented medical curriculum

Authors:

Pemra C Ünalan (pcunalan@gmail.com)
Arzu Uzuner (arzuuzuner@gmail.com)
Serap Çifcili (serapcifcili@gmail.com)
Mehmet Akman (makman4@gmail.com)
Sertaç Hanci (dr_sertach@hotmail.com)
Hans O Thulesius (hans.thulesius@ltkronoberg.se)

Version: 5 Date: 15 October 2009

Author's response to reviews:

Dear Editor,

Thank you for the open and fast revision of our manuscript. Feedback of the reviewers made important changes in it. However before acceptance some further changes are made parallel to your comments.

1- A "competing interest" section is added between the conclusion and authors' contribution sections.

2-An ethical approval statement of the study is clarified by the following explanation at the end of the method section; "Ethical approval for the study was not needed but an approval to conduct the study was obtained from the Educational Board of Coordinators of the Medical Faculty."

3-The first mentioned percentage under the "Final exam" subtopic of the results section is changed from 74% to 76%.

4-e-mail adress of the fifth author is changed as "dr_sertach@hotmail.com" both in the original manuscript downloaded and in the author details page of the online submission.

5-We also went through the manuscript formatting checklist one more time and made last revisions in the reference list section depending on the check-list.

With my best regards,

Pemra C.Unalan, MD.

No other change was done.

Thank you for the efforts to review the manuscript.